Author's response to reviews

Title: Evaluation of a complex intervention to improve Activities of Daily Living of disabled cancer patients: Protocol for a randomised controlled study and feasibility of recruitment and intervention.

Authors:

Line E Lindahl-Jacobsen (llindahl@health.sdu.dk)
Dorte G Hansen (dgilsaa@health.sdu.dk)
Karen la Cour (klacour@health.sdu.dk)
Jens Søndergaard (jsoendergaard@health.sdu.dk)

Version: 2
Date: 13 March 2014

Author's response to reviews:

Dear editor.

Please find enclosed the revised evrsion of our manuscript.

Kind regards,

Line Lindahl-Jacobsen